Author’s response to reviews

Title: Australian integrative oncology services: a mixed-method study exploring the views of cancer survivors.

Authors:

Jennifer Hunter (drjenniferhunter@yahoo.com.au)
Jane Ussher (J.Ussher@westernsydney.edu.au)
Chloe Parton (C.Parton@westernsydney.edu.au)
Andrew Kellett (A.Kellett@westernsydney.edu.au)
Caroline Smith (caroline.smith@westernsydney.edu.au)
Geoff Delaney (Geoff.Delaney@sswhs.nsw.gov.au)
Eleanor Oyston (eleanor@oncologymassagetraining.com.au)

Version: 4 Date: 14 Apr 2018

Author’s response to reviews:

Dear Editor

A title page has been uploaded

The COREQ checklist and manuscripts with track changes etc have been removed.

Single, final, clean versions of manuscripts and tables have been uploaded

kind regards

Jennifer Hunter